Author's response to reviews

Title: Procalcitonin is not sufficiently reliable to be the sole marker of neonatal sepsis of nosocomial origin

Authors:

Jose B Lopez Sastre (jblopez@uniovi.es)
David Perez Solis (david.perez@sespa.princast.es)
Vicente Roques Serradilla (roes5@ono.com)
Belen Fernandez Colomer (bcolomer@arrakis.es)
Gil Daniel Coto Cotallo (gildaniel.coto@sespa.princast.es)
Xavier Krauel Vidal (xkrauel@hsidbcn.org)
Eduardo Narbona Lopez (eduardo_narbona@hotmail.com)
Manuel Garcia del Rio (mgarcia@uma.es)
Manuel Sanchez Luna (msanchezl.hugm@salud.madrid.org)
Antonio Belaustegui Cueto (jose_lopez_sastre@hotmail.com)
Manuel Moro Serrano (mmoro.hcsc@salud.madrid.org)
Alfonso Urbon Artero (alfonsourbon@yahoo.es)
Emilio Alvaro Iglesias (med015929@nacom.es)
Angel Cotero Lavin (acotero@hruc.osakidetza.net)
Eduardo Martinez Villalta (jose_lopez_sastre@hotmail.com)
Bartolome Jimenez Cobos (jjimenez_bar@gva.es)

Version: 4 Date: 11 May 2006

Author's response to reviews:

We are delighted that our manuscript is going to be published by BMC Pediatrics. We have made the changes you requested for the production process:

- Word count on abstract's page: it has been removed.

- Author's contributions: it has been changed to Authors' contributions.

- References: we have included full authors list, removing the term "et al". We inform you that it was not a RefMan records' problem but a biomedcentral RefMan's output style included with the software.

Sincerely yours,

The authors